Author's response to reviews

Title: Metastatic ameloblastoma responding to combination chemotherapy: case report and review of the literature

Authors:

Mounia Amzerin (mounia.amzerin@gmail.com)
Zouhour Fadoukhair (zouhour.fa@hotmail.com)
Rhizlane Belbaraka (belbaraka@hotmail.com)
Meryem Iraqui (iraqi@hotmail.com)
Saber Boutayeb (saber@hotmail.com)
Hind M'rabeti (hind@hotmail.com)
tayeb kebdani (tayeb@hotmail.com)
khaled Hassouni (khaled@hotmail.com)
Najib Benjaafar (najib@hotmail.com)
Brahim K El gueddari (gueddari@hotmail.com)
Hassan Errihani (h_errihani@yahoo.fr)

Version: 5 Date: 23 August 2011

Author's response to reviews: see over
<table>
<thead>
<tr>
<th>Comment</th>
<th>proposed correction</th>
</tr>
</thead>
<tbody>
<tr>
<td>written English</td>
<td>I revised the manuscript with an English professor. So the grammatical and spelling</td>
</tr>
<tr>
<td></td>
<td>mistakes can be corrected without changing the meaning.</td>
</tr>
<tr>
<td>Keyword section</td>
<td>Removed</td>
</tr>
<tr>
<td>Competing interests section</td>
<td>Added: The author(s) declare that they have no competing interests</td>
</tr>
</tbody>
</table>